Author's response to reviews

Title: TIPIT: A Randomised Controlled Trial of Thyroxine in Preterm Infants under 28 weeks Gestation: Magnetic Resonance Imaging and Magnetic Resonance Angiography Protocol

Authors:

Sze M Ng (M.Ng@liverpool.ac.uk)
Mark A Turner (mark.turner@liverpool.ac.uk)
Carrol Gamble (C.gamble@liverpool.ac.uk)
Mohammed Didi (mohammed.didi@rlc.nhs.uk)
Suresh Victor (Suresh.victor@manchester.ac.uk)
Christina Malamateniou (Christina.Malamateniou@liverpool.ac.uk)
Laura M Parkes (laupar@liverpool.ac.uk)
Anna Tietze (anna.tietze@rlc.nhs.uk)
Lloyd Gregory (lloyd.gregory@manchester.ac.uk)
Vanessa Sluming (vanessa.sluming@liverpool.ac.uk)
Laurence Abernethy (laurence.abernethy@rlc.nhs.uk)
Alan Michael Weindling (a.m.weindling@liverpool.ac.uk)

Version: 6 Date: 26 June 2008

Author's response to reviews:

Formatting changes requested

-----------------------------

Major revisions (necessary)

Corresponding author - One or more corresponding author(s) must be indicated on the title page.
Action- This has been indicated in the title page

Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.
Action- As this is a Protocol, there is no "Results" and "Conclusions"

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).
Action- As this is a Protocol, there is no "Results", "Discussion" and "Conclusions"

Minor revisions (we can make these changes for you)
Title - please remove the full stop from the end of the title. Please ensure you also update the title in the submission system.
Action- Done

Author list - Please remove titles from the list of authors (e.g Dr / Professor) and include the authors' names as a paragraph, not a list.
Action- Done

Section headings - Please use sentence case for section headings i.e. Abstract not ABSTRACT.
Action- Done

BACKGROUND INFORMATION - Please rename this section 'Background'.
Action- Done

Sponsors - Please include this information in the Acknowledgements section.
Action- Done

Tables - Please move the tables to a position after the references.
Action- Done